ERIA Discussion Paper Series

Technology Transfer in ASEAN Countries: Some Evidence from Buyer-Provided Training Network Data^{*}

Fukunari KIMURA[†] Keio University and ERIA

Tomohiro MACHIKITA[‡] Institute of Developing Economies (IDE-JETRO)

Yasushi UEKI[§] Economic Research Institute for ASEAN and East Asia (ERIA) May 2015

Abstract: Technology transfers are important channels for firms in developing countries to get access to new technology and initiate innovation. This paper examines the geographical pattern of technology transfers in the form of buyer-provided training in domestic and international production networks. Our unique buver-supplier network data in four countries in Southeast Asia allow us to directly observe the buyer-supplier relationship as well as the existence of inter-firm provision of training for product/process innovation in order to investigate the geographical structure of knowledge acquisition, dissemination, and aggregation among local and non-local firms. The empirical analysis finds the following: (i) the probability of having training provided by the main buyer presents a U-shaped quadratic pattern with respect to the geographical distance between the respondent firms and the main buyers. The geographical proximity to the main buyer seems to be particularly important for local firms. (ii) The training provision is likely for both local and non-local firms when the main buyer is a multinational located in the same country. (iii) The probability of having training from the main buyer is high when the main buyer conducts R&D. (iv) Both local and non-local firms that have training provided by their main buyers are likely to provide training to their main suppliers. (v) In the case of non-local firms, product innovation with production partners is more likely when they have upstream/downstream training. However, such links seem to be weaker in the case of local firms.

Keywords: buyer-provided training; FDI spillovers; backward linkages; Southeast Asia *JEL Classification*: M5, O31, O32, R12.

^{*} The authors especially thank Yutaka Arimoto, Momoko Kawakami, Keiko Ito, Yoichi Sugita, Mari Tanaka, Willem Thorbecke, Kazunari Tsukada and Shujiro Urata for their insightful inputs, comments and seminar participants at IDE-JETRO and The East Asian Economic Association (EAEA), 2014 for helpful comments. This project could not have been carried out without cooperation from the Indonesian Institute of Sciences (LIPI) of Indonesia, the Philippine Institute for Development Studies (PIDS), the Sirindhorn International Institute of Technology (SIIT), Thammasat University of Thailand, the Institute for Industry Policy and Strategy (IPSI), Ministry of Industry and Trade of Viet Nam, Bangkok Research Center of IDE-JETRO, and Economic Research Institute for ASEAN and East Asia (ERIA) for financial supports to our questionnaire survey. Support for Machikita's and Ueki's research from JSPS (No. 21320212 and No. 25380559) is also gratefully acknowledged. We are solely responsible for any errors that may remain.

[†] Email: fkimura@econ.keio. ac. jp

[‡] Email: Tomohiro_Machikita@ide.go.jp

[§] Email: yasushi.ueki@eria.org

1. Introduction

Innovation is important for not only advanced countries but also developing countries. Although the common belief would be a complete vacuum of innovation activities in developing countries, careful micro-level observation proves that firms in developing countries also conduct process innovation and sometimes even product innovation though the depth and width of innovation may be at a primitive stage. The issue in developing countries is how such innovation can become pervasive among a large number of local firms and the quality of innovation can be upgraded in order to lead continuous productivity enhancement and sustained economic growth.

ASEAN (The Association of South-East Asian Nations) Member States have traced a unique growth trajectory. In contrast to the pattern of industrialization in Japan, Taiwan, and South Korea in the 1950s to 1980s where the development of indigenous firms and industries was at the center, ASEAN have aggressively utilized the mechanics of production networks extended by multinationals and have jump-started their industrialization processes. These countries have further proceeded to the formation of industrial agglomeration in parallel with the fragmentation of production where both "international" production networks and "domestic" vertical links of production have developed hand in hand. This new development strategy has proved to be successful in accelerating industrialization. As for the development of international production networks in ASEAN and East Asia, see Ando and Kimura (2005). The concept of two-dimensional fragmentation of production proposed by Kimura and Ando (2005) is also useful for understanding the parallel development of fragmentation and agglomeration.

Innovation should be explored in the particular context for ASEAN and other developing countries that utilize the mechanics of production networks. Table 1 presents some basic statistics of ASEAN and other Asian countries. Four countries in our study, Thailand, Indonesia, the Philippines, and Viet Nam, belong to the upper or lower middle income level ranging from US\$1,755 to US\$5,480 per capita, and average years of schooling in these countries indicate their steady bottom-ups of education. However, we have to note that R&D expenses are very small even though the ratios of manufacturing value added to GDP show extensive industrialization. This

is because their industrialization has heavily depended on production networks by multinationals and local research capability has not been developed much. Inward FDI (foreign direct investment) stocks to GDP ratios are indeed high, particularly in Thailand (48 percent) and Viet Nam (48 percent). In the catching-up process, countries do not have to explore absolutely new technology and scientific knowledge. Instead, they must utilize the advantage of latecomers and effectively capture technology transfers or spillovers in order to expand and upgrade innovation. Given the current low local capability of absorbing and disseminating technologies, the issue is whether international and domestic production networks can be vehicles for technology transfers from developed to developing economies and from multinationals to local firms.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Variables	GDP per capita (US\$)	Average years of schooling	R&D expenses	Mfg value added	Royalty receipts	Royalty payments	Inward FDI stock
		e	(% of	(% of	(% of	(% of	(% of
			GDP)	GDP)	GDP)	GDP)	GDP)
Year	2012	2010	2009	2010	2012	2012	2013
Singapore	51709.45	10.63	2.20%	21.63%	0.57%	5.75%	283.20%
Japan	46720.36	11.52	3.36%	19.65%	0.54%	0.33%	3.50%
South Korea	22590.16	11.89	3.56%	30.29%	0.28%	0.69%	13.70%
Malaysia	10432.06	9.75	1.01%	24.52%	0.04%	0.50%	46.30%
China	6091.01	7.12	1.70%	32.46%	0.02%	0.24%	10.40%
Thailand	5479.76	7.30	0.25%	35.62%	0.07%	0.99%	47.90%
Indonesia	3556.79	7.26	0.08%	24.80%	0.01%	0.21%	26.50%
The Philippines	2587.02	8.18	0.11%	21.44%	0.00%	0.20%	12.00%
Viet Nam	1755.21	7.45		17.95%	N.A.	N.A.	47.90%
India	1489.23	5.39	0.82%	14.87%	0.02%	0.21%	12.10%
Lao PDR	1417.08			7.47%	N.A.	N.A.	27.80%
Cambodia	944.41	4.10		15.62%	0.03%	0.08%	60.00%

Table 1: Comparisons of Technology-level between East Asia and ASEAN countries

- *Notes*: (1) GDP per capita is gross domestic product divided by midyear population. Data are in current U.S. dollars. R&D expenses (% of GDP) are expenditures for R&D activities divided by GDP. Mfg value added (% of GDP) is the net output of manufacturing sector after adding up all outputs and subtracting intermediate. Royalty receipts (% of GDP) and payments (% of GDP) are charges for the use of intellectual property. These are not applicable (N.A.) for Viet Nam and Lao PDR. Inward FDI stock (% of GDP) is stock of inward foreign direct investment divided by GDP.
- Sources: World Bank national accounts data 2014 for GDP per capita. Barro-Lee Educational Attainment Dataset for average years of schooling. Science, Technology, and Innovation Dataset of UNESCO for R&D expenses (% of GDP). World Bank national accounts data

2014 for Mfg value added (% of GDP). World Development Indicators 2013, World Bank for royalty receipts (% of GDP) and royalty payments (% of GDP). World Investment Report, UNCTAD 2014 for inward FDI stock (% of GDP).

Technology transfers can be achieved based on the technological capacity of local firms and their effort of knowledge acquirement. Knowledge may be acquired through a variety of channels; it would be done through learning itself, the introduction of foreign technology possibly by importing new capital goods or capturing unintentional technology spillover, interactions with foreign affiliates established by incoming FDI, learning from export, and others. In international and domestic production networks, we often observe buyer-provided training, which may be one of the important channels for intentional technology transfers in order to accelerate process innovation as well as possibly product innovation. In this paper, we would like to investigate how extensively buyer-provided training for process/product innovation is conducted, what would be the relationship with incoming FDI and export by looking at geographical structure of such training, and how far such knowledge acquiring effort would be spread through production networks.

There is a long list of previous studies on knowledge diffusion among enterprises. Keller (2004, 2012) provides excellent overviews of technology transfers across countries. These surveys present economic theories and empirical evidences on multiple ways of costly inter-firm learning through FDI, exporting, and importing. Among several channels of technology transfers, vertical linkages have been specifically tested in a series of recent empirical papers as a knowledge transmission mechanism between upstream and downstream firms, in the context of developing and emerging economies. For example, Aitken and Harrison (1999) show positive impacts of foreign equity participation on plant productivity of small enterprises by using Venezuelan plant-level data. They also find that foreign investment negatively affects the productivity of domestic plants through enhancing competition. They conclude that a net gain from FDI is quite small if we take into account the two offsetting effects. On the other hand, Javorcik (2004) and Blalock and Gertler (2008) find backward linkage impacts from multinational enterprise (MNE) customers on local suppliers' productivity growth by using the share of MNEs in downstream sectors as an explanatory variable. Blalock and Gertler claim the existence of sizable technology transfers from downstream MNEs to upstream firms, based on empirical estimates of the relationship between the share of MNEs in downstream firms and productivity growth in upstream firms. One of the shortfalls of the past literature, however, is to depend only on indirect evidences of technology transfers, rather than pinpointing what sort of technological information is transferred from whom to whom. Another shortage of the past literature is to lay too much emphasis on North-South technology transfer. South-South FDIs and trades among East Asian developing economies are emerging technology transfer channels in ASEAN and East Asia. Machikita and Ueki (2013) investigate technology transfer from China and Japan to Viet Nam.

To fill these gaps, we collected unique buyer-supplier linked data in four Southeast Asian countries to investigate the geographical structure of technology transfers, particularly in the form of buyer-provided training, in domestic and international production networks. One of the novelties in our study is to directly observe links of technology transfers. Previous representative studies such as Javorcik (2004) and Blalock and Gertler (2008) have depended on vertical linkage information based on input-output tables in order to postulate possible horizontal and vertical linkages, not directly capturing actual links. An important exception is Gorodnichenko, *et al.* (2010) who utilize information on direct linkages between domestic and foreign firms to identify several channels of globalization impacts. In the setting limited to agriculture, Conley and Udry (2010) provide a framework and evidence on what sort of neighbors affects productivity growth for pineapple farmers in Ghana. We believe that direct observation of technology transfers in the form of training reveals the nature of vertical linkages among firms including multinationals and local firms located nearby and abroad.

Another unique feature of our study is to focus on buyer-provided training as a particular channel of vertical technology transfers. One attribute of training is to enhance human capital at the individual level, and the literature seems to be interested in the implication of labor turnovers in the globalizing environment.¹ However, in this

¹ Gershenberg (1987) claims that managerial knowledge rarely flows from multinationals to domestic firms in Kenya. Gorg and Strobl (2005) find that enterprise owners who used to work for multinationals move to more productive firms than those who worked only for domestic firms by using the survey from Ghana. Markusen and Trofimenko (2009) show that foreign experts can transfer the knowledge to domestic workers in Colombia. Balsvik (2011) presents that workers with MNE experience have a more sizable contribution to the productivity of their plants than workers without MNE experience in the case of Norwegian manufacturing sector. Poole (2013) documents wage spillovers within a firm through hiring well-trained workers from multinationals

study, training is interpreted as an effort for facilitating and upgrading vertical links in production networks. Not all firms are automatically qualified as participants in production networks. There is a strict qualifying selection, and after that, buyers may have an incentive to provide supplementary training for suppliers in order to teach how to introduce new products, improve production processes, and ultimately make production networks work well.² When downstream production partners have global market access and advanced knowledge, they may set out to share their technical knowledge of new products and production processes with their upstream suppliers. Such learning may be linked further along the production networks; i.e., information recipients could become information senders.

The major findings are summarized as follows: first, the probability of having buyer-provided training shows a U-shaped quadratic pattern with respect to the geographical distance between the respondent firms and the main buyers. In other words, the probability is high when the distance is either short or long. It is particularly important for local firms to locate nearby the main buyer to have buyer-provided training not across borders. Second, training provision is likely for both local and non-local firms when the main buyer is a multinational located in the same country. Non-local firms also tend to have training when the main buyer is located in ASEAN, East Asia, or the rest of the world. Foreign buyers play an important role for buyer-provided training to non-local exporters in ASEAN, while such spillovers through exporting do not work for local firms. Third, the probability of having training from the main buyer is high when the main buyer conducts R&D while the capital tie with the main buyer does not seem to enhance the probability.³ These results imply that training provision is expected to spur by main buyers who have high technological levels. Fourth, both

located in Brazil, but the extent of wage spillovers is limited to higher-skilled domestic workers, not lower-skilled workers. Stoyanov and Zubanov (2012) find that Danish firms hiring workers from more productive firms can experience productivity gains and these gains increase with skill-level of new workers.

 $^{^{2}}$ As Antras and Rossi-Hansberg (2009) summarizes, Jones and Kierzkowski (1990) provide a theoretical framework on observed fragmentation of production, explaining a tradeoff between gains from fragmentation, when the economic advantages of cross-border production are utilized, and the costs of disintegration from cross-border production.

³ This finding on the capital relationship between buyers and suppliers seems to be different from the implication of Atalay, *et al.*, (2014) which emphasized the importance of intangible inputs. They concluded that the prime motivation for owning multi-plants is to make more efficient transfers of knowledge of production and information on markets.

local and non-local firms that have training provided by their main buyers are likely to provide training to their main suppliers. An information recipient from the downstream buyer is likely to be an information sender to its upstream supplier along the production chain; this suggests that production networks within and across borders seem to work as a chain of technology transfers. Finally, in the case of non-local firms, product innovation with production partners is more likely when they are information recipients/senders in terms of training. This suggests a tight link of innovation with technology acquisition, knowledge dissemination, and information aggregation for non-local firms, but such links seem to be weak for local firms.⁴ The scope of this study is limited in the sense that it concentrates on one particular form of technology transfers, i.e., buyer-provided training. However, we believe that it is worthwhile analyzing the geographical structure of technology transfers in domestic and international production networks with direct observation.

The remaining parts of this paper are organized as follows: section 2 introduces our dataset on firms' self-reported main buyers and main suppliers. It provides the evidence for technology transfers in the form of buyer-provided training in the global value chain network. Section 3 presents the geographical formation of buyer-provided training in a global supply chain. We also demonstrate technology transfers from downstream to upstream firms. Section 4 provides the evidence of product innovation that is correlated with incoming and outgoing technology transfers. Section 5 concludes the paper.

⁴ Our findings are partially supported by qualitative case studies on how buyer-supplier trade relationships solve the problem associated with transfers of tacit knowledge, particularly through buyers' provision of technical assistance to their suppliers. An example is found in the transport equipment industry including automobiles and motorcycles in Viet Nam. Viet Nam Manufacturing and Export Processing Co., Ltd. (hereafter VMEP), a motorcycle manufacturer, occasionally assigns technical officers to provide technical assistance for its suppliers. In its close collaborative relationship with Chiu Yi Viet Nam (hereafter CYV), VMEP usually appoints technical staffs in order to enable CYV to enhance product quality as well as supporting the development of new products and designs. Meanwhile, CYV appoints technical staffs to examine process errors on the spot or take part in meetings about quality enhancement as well as submitting a proposal to VMEP on improving technology and product styles (Truong Chi Binh, 2010). A number of case studies have also revealed that the transportation equipment sector is dominated by MNE assemblers who provide their suppliers technical assistance in return for high quality control on the supplier side. We can say that being a part of MNEs' production network is just like attending a "training school" for its suppliers (Intarakumnerd, 2010).

2. The Firm's Self-reported Buyer and Supplier Data

2.1. The Survey

This paper employs the questionnaire survey data that we collected from firm managers and correspondences. The sample population is restricted to manufacturing firms operating in main industrial districts in four ASEAN countries—the JA-BODETABEK area in Indonesia that comprises Jakarta, Bogor, Depok, Tangerang, and Bekasi; the CALABARZON area in the Philippines that consists of Cavite, Laguna, Batangas, Rizal, and Quezon; the Greater Bangkok area in Thailand; and the Hanoi and Ho Chi Minh City areas in Viet Nam.

Our questionnaire starts by asking a respondent establishment on its basic characteristics such as the year of establishment, capital structure (100 percent locally owned, hereafter "local"), 100 percent foreign owned (MNE), or joint venture (JV)), and size (in terms of employees and assets). Then it tries to quantify achievements in product and process improvements and sources of technologies and information used by the establishment to conduct innovative activities. It also asks questions regarding a main buyer and supplier for the respondent establishment which include the attributes of its main buyer (supplier), collaborative relationships between the respondent and its main buyer (supplier), the distance between them, and the country where the main buyer (supplier) locates.

The survey began in November 2012 and was completed in January 2013. Responses were collected by mail, phone interviews, and face-to-face interviews. 979 establishments agreed to participate in the survey: 157 in Indonesia (16 percent of the total), 218 in the Philippines (22 percent), 284 in Thailand (29 percent), and 320 in Viet Nam (33 percent). The original survey shows that 55 percent of the responded establishments are 100 percent locally owned, and 53 percent of them are SMEs hiring less than 200 employees.

2.2. Characteristics of the Respondent Firms

From the 979 raw observations, we can use 807 restricted observations for this paper as summarized in Table 2. The decrease in the number of observations is caused mainly by missing values for the respondents in Thailand. Consequently about 19.3

percent of the respondents locate in Indonesia, 26.5 percent in the Philippines, 15.1 percent in Thailand, and 39.0 percent in Viet Nam. However, this decrease does not make a significant difference in the importance of local firms and SMEs between the raw and restricted datasets. By the capital structure, 52.4 percent of the 807 respondents are local firms. Multinational enterprises and joint ventures are 36.2 percent and 11.4 percent of the 807 respondents, respectively.

As shown in Table 2, the main manufacturing activity of the respondents is concentrated in machinery (23.4 percent of the sample), chemical products (17.8 percent), automobiles and other transport equipment (11.3 percent), textiles (11.9 percent), and metal products (11.9 percent), and the industrial composition of the dataset for each country is not widely different from that of official statistics in each country (Machikita and Ueki, 2015). The average age of the respondents is 17.2 years. They mainly produce parts and components (33.1 percent) or assemble final goods (48.6 percent). 36.8 percent of them are OEM (original equipment manufacturer).

With respect to the establishment size, the respondents choose one of the following eleven categories: 1-19 persons, 20-49, 50-99, 100-199, 200-299, 300-399, 400-499, 500-999, 1,000-1,499, 1,500-1,999, and 2,000 and more. The variable for the number of employees is defined as the median value of each category, of which the mean is 397.5 as shown in Table 2.

The questionnaire asks the respondents whether they conduct R&D and have achieved process improvements. We use R&D as indicators for absorptive capability (Cohen and Levinthal, 1990). Table 2 shows 51.9 percent of the respondents conduct R&D. It also introduces an indicator for innovative capability that is defined as the standardized sum of 12 dummy variables for process improvements, which have a mean of zero and a standard deviation of one, as defined by Machikita and Ueki (2015) which applies the methods by Ichniowski, *et al.* (1997) and Bloom and van Reenen (2007) to manufacturing firms in Southeast Asia.

	Whole	N=807 Std. Dev	Local	N=423 Std. Dev	Non-local Mean	N=384 Std
	wiedi	Stu. Dev.	Ivican	Stu. Dev.	Ivicali	Dev.
Main buyer characteristics						
Main buyer gave training	-					
Gave training to main supplier	0.374	0.484	0.272	0.445	0.487	0.5
Develop a new product with	0.297	0.457	0.201	0.401	0.404	0.491
production partners Design a	0.26	0.439	0.189	0 392	0 339	0 474
new product with main buyer	0.379	0.485	0.305	0.461	0.355	0.499
Distance to main buyer (km)	557 807	803 150	341 625	605.065	706 133	010 530
Domestic local buyer	0 405	0 /01	0.61	0.488	0.18	0 384
Domestic MNF buyer	0.405	0.491	0.01	0.400	0.18	0.384
Domestic IV huver	0.135	0.370	0.132	0.339	0.138	0.455
Foreign buyer in ASEAN	0.133	0.215	0.028	0.166	0.150	0.256
Foreign buyer in East Asia	0.128	0.334	0.028	0.100	0.227	0.230
Foreign buyer in EU or US	0.062	0.241	0.043	0.202	0.083	0.277
Foreign buyer in ROW	0.012	0.111	0.014	0.118	0.01	0.102
Capital tie with main buyer	0.229	0.421	0.189	0.392	0.273	0.446
Number of employees of main	497.638	381.408	400.936	364.574	604.162	371.462
Main buyer is doing R&D	0.441	0.497	0.392	0.489	0.495	0.501
Main supplier characteristics						
Distance to main supplier (km)	728 538	853 154	179 793	689.07	1002 546	930.04/
Domestic local supplier	0 322	0 468	0 511	05	0 115	0 319
Domestic MNE supplier	0.107	0.309	0.104	0.306	0.109	0.313
Domestic IV supplier	0 141	0 349	0.142	0 349	0 141	0.348
Foreign supplier in ASEAN	0.067	0.25	0.035	0.185	0.102	0.302
Foreign supplier in East Asia	0.289	0.453	0.149	0.356	0.443	0.497
Foreign supplier in EU or US	0.043	0.204	0.028	0.166	0.06	0.238
Foreign supplier in ROW	0.019	0.135	0.012	0.108	0.026	0.159
Capital tie with main supplier	0.218	0.413	0.196	0.398	0.242	0.429
Number of employees of main	364.921	312.930	311.575	302.938	423.685	313.617
Main supplier is doing R&D	0.315	0.465	0.305	0.461	0.326	0.469
Respondent firm characteristics						
Capability (z-score)	0	1	-0.025	1.01	0.027	0.99
Doing R&D	0.519	0.5	0.515	0.5	0.523	0.5
OEM	0.368	0.483	0.364	0.482	0.372	0.484
Firm age	17.232	14.287	20.489	16.444	13.643	10.345
Number of employees	397.497	524.739	302.896	471.684	501.706	559.952
Local	0.524	0.500	1	0	0	0
Multinational enterprises	0.362	0.481	0	0	0.76	0.427
Joint ventures	0.114	0.318	0	0	0.24	0.427
Parts and components	0.332	0.471	0.203	0.403	0.474	0.5
Final goods production	0.486	0.5	0.608	0.489	0.352	0.478
Indonesia	0.193	0.395	0.251	0.434	0.130	0.337
The Philippines	0.265	0.442	0.236	0.425	0.297	0.457
I hailand	0.151	0.358	0.232	0.422	0.063	0.242
viet Nam	0.390	0.488	0.281	0.450	0.510	0.501
rood Tautila	0.09/	0.296	0.13/	0.344	0.052	0.222
Textile Paper	0.119	0.324	0.139	0.34/	0.096	0.295
r aper	0.048	0.213	0.009	0.233	0.020	0.139
Uneinicai Matal	0.178	0.383	0.177	0.382	0.18	0.384
Machinery	0.119	0.524	0.120	0.334	0.109	0.513
	0.234	0.424	0.105	0.370	0.515	0.404
Auto	0.115	0.510	0.05	0.217	0.102	0.307

Table 2: Summary Statistics for Local and Non-local Firms

Note: Local is 100 percent local capital firms. Non-local is 100 percent foreign capital firms (i.e., multinational enterprises) and joint venture firms. Source: ERIA Establishment Survey 2012.

2.3. Characteristics of a Firm's Main Buyer and Supplier

Table 2 also reports summary statistics for a number of variables regarding the main buyer and the main supplier. There are two types of variables for the distance between the respondent and its main buyer or supplier. One is the distance expressed in kilometers. The questionnaire asks a respondent to indicate the distance to its main buyer (supplier) by choosing one of the following eleven categories: 0-10 kilometers, 11-25, 26-50, 51-100, 101-200, 201-300, 301-400, 401-500, 501-1,000, 1,001-2,000, and more than 2,000. The variable for the distance to the main buyer or supplier is defined as the median value of each category. Table 2 indicates the means of the distance to the main buyer and supplier are 557.9 and 728.5 kilometers respectively. This variable for kilometer distance is log-transformed and squared for the regression analyses in the following sections.

The other variables for the distance are dummy variables defined by the location where the main buyer or supplier resides. We categorized its location into the following five regions: domestic, ASEAN, East Asia (China, Japan, Korea, and Taiwan), EU or US, and the rest of the world (ROW). If the main buyer or supplier is domestic, in other words, located in the same country as the respondent, the main buyer/supplier is further categorized according to its capital structure (local, MNE, JV). Table 2 indicates that many respondents have their main buyers or suppliers domestically, particularly when respondents are local. However, a considerable number of the respondents have their main buyers or suppliers in East Asia while a non-negligible portion has them in ASEAN. EU and the US are still important for ASEAN as buyers, in other words, as export destinations.

There are other attributes of the main buyer or supplier reported in Table 2 that may affect technology transfers between the respondent and its main buyer or supplier. 22.9 percent and 21.8 percent of the respondents have capital ties with their main buyers and suppliers respectively. 44.1 percent and 31.5 percent have main buyers and suppliers who conduct R&D. The means of the number of employees for the main buyers and suppliers are 497.6 and 364.9 respectively. These variables for the number of employees are defined in the same way as the size of the respondents; the survey requested the respondent to indicate the number of employees hired by its main buyer (supplier) by choosing one of the following five categories: 99 or less employees, 100199, 200-299, 300-999, and 1,000 and more.

2.4. Main Dependent Variables

Main dependent variables in our regression analyses are dummy variables for technology transfers through the provision of training by a downstream firm to its upstream supplier. Although the questionnaire does not specify the detailed contents of "training," it certainly intends to help the counterparts introduce new intermediate products, economize production processes, and make production networks work well. The unique data on the buyer-supplier relationship allow us to develop two variables related to such technical assistance. One is the training provided for the respondent by its main buyer: the dummy variable is coded 1 if the buyer provides training to the respondent for its main supplier: the dummy variable is coded 1 if the respondent provides training to its main supplier; otherwise 0. Table 2 presents 37.4 percent and 29.7 percent of the respondents are provided training by their main buyers and provide training to their main supplier respectively. We regress these dummy variables for training provision on the distance variables and other independent and control variables.

This paper further asks whether the training would link to the formation of collaborative buyer-supplier relationships in order to realize innovations. There are two dummy variables for the collaboration for product development. One is the dummy variable that equals 1 if the respondent has introduced a new product in cooperation with other firms. The other dummy variable is codified 1 if the respondent designs a new product with its main supplier, otherwise 0. As shown in Table 2, 26.0 percent of the respondents have introduced a new product in cooperation with other firms, whereas 38.0 percent of them design a new product with their main suppliers.

3. Regression Results for Buyer-provided Training

This section presents empirical results on buyer-provided (sponsored) technology transfers to suppliers in the form of training provision. First, we check whether buyer-

provided (sponsored) technology transfers within a chain are penalized by physical distance between the suppliers and the buyers. Then we investigate whether FDI in developing economies plays an important role for technology transfers through buyer-provided training to local suppliers, in comparison with other possible technology transfer/spillover channels such as spillovers from local buyers and spillovers through exporting. We will see how international exposure of suppliers (i.e., having incoming FDI and exporting) affects technology transfers. Second, we test whether suppliers who receive buyer-provided training also provide training to their upstream suppliers. This is the test that checks how far information recipients can also be information senders in the backward linkage.

3.1. Incoming Technology Transfers shaped by Geography, FDI, and Exports

Our testable hypotheses are (1) the incidence of buyer-provided training to suppliers is negatively correlated with geographic distance between buyers (information senders) and suppliers (information recipients), and (2) the incidence of buyer-provided training to suppliers is positively correlated with having MNE buyers and exporting products to ASEAN buyers. To derive the empirical results, this paper estimates a simple model of technology transfers on buyer-seller networks as follows:

 $\begin{aligned} \Pr(Buyer \ gave \ training = 1)_i \\ &= \alpha + \beta_1 log(Distance \ to \ buyer)_i + \beta_2 log(Distance \ to \ buyer)_i^2 \\ &+ \beta_B(Buyer's \ covariates)_i + \beta_X(covariates)_i + \varepsilon_i \quad (1) \end{aligned}$

where dependent variable (*Buyer gave training*)_i signifies a binary variable which is equal to one if firm *i*'s main buyer gave training to respondents, otherwise zero. (*Distance to buyer*)_i denotes a distance between main buyer and respondent firms measured in kilometers. (*Buyer's covariates*)_i is capturing the main buyer's characteristics explaining outgoing technology transfers to upstream supplier firms, which include a dummy variable capturing capital ties between main buyers and respondents, main buyers' employment size, and a dummy variable capturing main buyers' R&D activities. (*covariates*)_i consists of a vector of firm-level observable characteristics capturing firm *i*'s capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, and industry dummies as well as country dummies. ε_i is an unobserved error.

Then we move on to estimating the following equation of technology transfers in a buyer-seller network:

 $\begin{aligned} Pr(Buyer \ gave \ training = 1)_i \\ &= \alpha + \gamma_1(Domestic \ MNE \ buyer)_i + \gamma_2(Foreign \ buyer)_i \\ &+ \gamma_B(Buyer's \ covariates)_i + \gamma_X(covariates)_i + \epsilon_i \end{aligned} (2)$

where (*Domestic MNE buyer*)_i is a binary variable which is equal to one if firm *i*'s main buyer is an MNE in each country, otherwise zero, (*Foreign buyer*)_i is also a dummy variable which is equal to one if firm *i*'s main buyer is in foreign countries, otherwise zero, and ϵ_i is an unobserved error. As we mentioned before, our hypothesis is that the incidence of incoming technology transfers from buyers is driven not only by geographic distance but also by multinational buyers in the same country through FDI and buyers abroad as export destination.

We provide a first look at the average adjusted prediction (AAP) computed from the first model. Although the AAPs are computed for the log-transformed distances, we plot them against the distance corresponding to each of the log-transformed distance. We present the figures for local and non-local firms separately to visualize different distance effects. Then we show the marginal impacts of geographic distance to the main buyer as well as the firm nationality and location of the main buyer estimated from the first and second models.

Let us start graphing AAP for local firms. Figures 1 plots the average adjusted prediction of the distance between firm *i* and firm *i*'s main buyer on the probability that local firm *i* receives buyer-provided training with 95 percent confidence intervals. It shows that the probability of having technology transfer through buyer-provided training to local firms in ASEAN follows a U-shaped quadratic pattern with respect to the physical distance to the main buyer. The average probability of receiving buyer-provided training for local firms starts from a bit lower than 0.45. The bottom of the U-shaped curve is around 200-250km to the main buyer where the probability of receiving buyer-provided training is 0.22. Then the upward sloping portion of the quadratic curve continues to 2,000km distance to the main buyer where the average probability of receiving buyer-provided training reaches close to 0.30.

How about the cases of non-local firms, i.e., multinational enterprises or joint ventures? Figure 2 also plots the average adjusted prediction of distance to main buyer on the probability that non-local firms *i* receives buyer-provided training. It clearly shows that technology transfer through buyer-provided training to non-local firms in ASEAN also present a U-shaped quadratic pattern with respect to the distance to the main buyer. The average probability of receiving buyer-provided training by non-local firms starts from the level of more than 0.75. The bottom of the U-shaped curve is again in around 200-250km distance to the main buyer where the probability of receiving buyer-provided training is 0.30. The upward sloping portion of the U-shaped curve for non-local firms follows a higher trajectory than local firms depicted in Figure 1, continuing to 2000km distance to the main buyer where the average probability of receiving buyer-provided training is close to 0.50. In sum, the comparison between Figures 1 and 2 provides the following intuitive findings: (1) Non-local firms are more likely to receive buyer-provided training than local firms; (2) Local firms receives training mainly from buyers located nearby in the same industrial agglomeration, and the role of foreign buyers are limited; (3) On the other hand, non-local firms are more likely to receive training from both buyers nearby and buyers abroad.

Figure 1: Plotting Average adjusted Prediction of Distance (km) to Main Buyer on the receiving Probability of buyer-provided training for Local Firms with 95 percent Confidence Intervals



Source: ERIA Establishment Survey 2012.

Figure 2: Plotting average adjusted prediction of distance (km) to main buyer on the receiving probability of buyer-provided training for non-local firms with 95 percent confidence intervals



Source: ERIA Establishment Survey 2012.

Regression results in Table 3 clearly indicate the pattern of backward linkages of technology transfers from downstream buyers to upstream suppliers. First, Columns 1 to 3 of Table 3 show that a U-shaped quadratic pattern with respect to the distance between firm i and firm i's main buyer exists on the decision of buyer-provided training to firm i. The quadratic pattern is significant for the whole sample (Column 1), the local sample (Column 2), and the non-local sample (Column 3). Embodied knowledge spillovers through training provided by downstream buyers are bimodal in terms of geographical distance.

Second, the capital tie-up between firm i and i's main buyer does not show any positive impact on incoming technology transfers from the buyer to firm i (Columns 1 to 3). In addition, non-local supplier i is more likely to experience buyer-provided training if the firm size of the main buyer is larger (Column 3), and both local and non-local firm i are more likely to have buyer-provided training if the main buyer conducts R&D (Columns 1 to 3).

Third, Columns 4 to 6 of Table 3 intend to identify technological transfers from buyers to suppliers in a production chain through FDI or exports. The reference group is local buyers. The positive coefficient of domestic MNE buyer dummy with a 1 percent statistical significance indicates that the probability of buyer-provided training to suppliers is high if the main buyer is MNE established by incoming FDI, compared with the situation that the main buyer is a local firm. This is true no matter whether firm i is a local or non-local supplier (Columns 5 and 6). This is a solid evidence of technological transfers through FDI to local suppliers (Columns 5) as well as technological transfers among non-local suppliers in each country. We cannot find clear results for the case that the main buyer is a joint venture.

Fourth, Columns 4 to 6 of Table 3 also seek the evidence of technology transfers from foreign buyers. We cannot find clear-cut results that foreign buyers would provide training to local suppliers (Column 5). In particular, foreign buyers are less likely to provide training to suppliers than local buyers in each country if foreign buyers are located in ASEAN or EU or US. On the contrary, foreign buyers located in any other ASEAN countries, East Asia, or the rest of the world seem more likely to provide training for non-local supplier *i* than local buyers (Column 6).

In summary, buyer-supplier network data and Table 3 demonstrate the following concrete evidences on technology transfers from buyers to suppliers: (1) the probability of technology transfers in the form of buyer-provided training for suppliers presents a U-shaped pattern with respect to the geographical distance between the buyer and the supplier; in particular, local buyers heavily rely on training provision in a short distance, (2) there exist technology transfers from downstream affiliates of multinationals located in the country to local suppliers while technology transfers from export-destination firms abroad to local suppliers seem to be weak, and (3) there are also significant technology transfers from downstream affiliates of multinationals located in the country to non-local suppliers, and technology transfers from export destination firms seem to work only for non-local suppliers in developing economies.

	(1)	(2)	(3)	(4)	(5)	(6)
Respondent firms in ASEAN	Whole	Local	Non-local	Whole	Local	Non-local
Distance to buyer						
Log of distance to main buyer (km)	-0.335***	-0.174***	-0.528***			
Square of log of distance to main	(0.056) 0.033*** (0.006)	(0.064) 0.016** (0.007)	(0.103) 0.052*** (0.010)			
Domestic vs. foreign buyer	(,	(,				
Domestic MNE buyer				0.319***	0.233**	0.447***
Domestic JV buyer				(0.058) -0.020	(0.105) -0.047	(0.075) 0.119
Foreign buyer in ASEAN				(0.059) 0.301***	(0.057) -0.167**	(0.111) 0.531^{***}
Foreign buyer in East Asia				(0.093) 0.065 (0.070)	(0.074) 0.208 (0.166)	(0.053) 0.233^{**} (0.007)
Foreign buyer in EU or US				(0.070) -0.116 (0.074)	-0.166***	(0.037) 0.039 (0.142)
Foreign buyer in rest of the world				(0.074) 0.005 (0.194)	-0.056 (0.157)	0.420*** (0.109)
Buyer characteristics				(0.25.1)	(******)	(01207)
Capital tie with main buyer	-0.096**	-0.074	-0.095	-0.115**	-0.056	-0.147**
Log of firm size of main buyer	(0.046) 0.075***	(0.062) 0.027	(0.067) 0.141***	(0.045) 0.064***	(0.064) 0.031	(0.065) 0.099**
Main buyer is doing R&D	(0.023) 0.145***	(0.025) 0.126**	(0.039) 0.131**	(0.023) 0.186***	(0.025) 0.145***	(0.039) 0.253***
Ν	(0.040) 807	(0.049) 423	(0.063) 384	(0.040) 807	(0.051) 423	(0.061) 384

Table 3: Impacts of the Distance from Main Buyers on Incoming Technology Transfers (training) from buyers

Note: All dependent variables are binary variables indicating whether the main buyer gave training to the respondent firm in each country in ASEAN. The reference group is domestic local buyers. Other control variables are a capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, industry controls (seven manufacturing industry with reference industry), and country controls.

Source: ERIA Establishment Survey 2012.

3.2. Impacts of Incoming Buyer-provided Technology Transfers on Outgoing Technology Transfers to Suppliers

Let us turn to investigate outgoing technology transfers from suppliers to suppliers' main suppliers. Our hypothesis here is that supplier firm *i* in Indonesia, the Philippines, Thailand, or Viet Nam is likely to provide training to its upstream supplier if firm *i* has been provided training by its down-stream buyer. The test result uncovers the supplier's chain behavior of diffusing knowledge from downstream to upstream in a production chain. To derive empirical estimates of the relationship between incoming and outgoing technology transfers, we estimate the following equation on the buyer-

supplier network:

 $\begin{aligned} Pr(Give \ training \ to \ supplier = 1)_i \\ &= \alpha + \mu_1(Buyer \ give \ training \)_i + \mu_S(Supplier's \ covariates)_i \\ &+ \mu_B(Buyer's \ covariates)_i + \mu_X covariates)_i + u_i \end{aligned} (3)$

where variable (*Give training to supplier*)_{*i*} is a dummy variable which is equal to one if supplier firm *i* provides training to its upstream supplier, otherwise zero, and (Buyer gave training)_i signifies a dummy variable which is equal to one if firm i's main buyer provides training to firm *i*, otherwise zero. (Supplier's covariates)_{*i*} captures main supplier's characteristics which includes the distance between firm i and its main supplier, a set of dummy variables of the main supplier's types (domestic or foreign supplier), a dummy variable capturing a capital tie-up between the main supplier and firm *i*, the main supplier's employment size, and a dummy variable for the main supplier's R&D activities. (Buyer's covariates)_i is also for the main buyer's characteristics that explain technology transfers to firm *i*, including a dummy variable capturing a capital tie between the main buyer and firm *i*, the main buyer's employment size, and a dummy variable for the main buyer's R&D activities. $(covariates)_i$ is a vector of firm-level observable characteristics which consist of firm i's capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, and industry dummies as well as country dummies. u_i is an unobserved error.

Table 4 presents the backward knowledge diffusion of three parties from downstream buyers and the respondents to their suppliers in a production chain through buyer-provided training. Columns 1 of Table 4 shows impacts of incoming technology transfer from firm i's main buyer to firm i on outgoing technology transfers from firm i's main supplier, after controlling the distance to the main supplier and the main buyer in the quadratic form. There are significant impacts of receiving buyer-provided training on providing training to the upstream supplier. Columns 2 and 3 of Table 4 present that impacts of incoming technology transfers on outgoing technology transfers are statistically significant for both local and non-local respondents.

The Columns 4 to 6 of Table 4 control the detailed relationship among firm *i*, firm

i's main buyer, and firm *i*'s main supplier. These results confirm the baseline results of Columns 1 to 3, which go with the geographic distances (km) among these three parties; impacts of receiving training from the downstream buyer on providing training to the upstream supplier are again statistically significant at 1 percent for the whole, local, and non-local respondents. Local firms also become information senders by providing training to their supplier if they are recipients of buyer-provided training. It is particularly true for non-local firms. Furthermore, non-local firms are more likely to provide training to their supplier if they have an MNE buyer in each country. In addition, non-local firms are more likely to provide training to their supplier if they suppliers in each country or they import intermediate goods from the JV suppliers in ASEAN.

In summary, Table 4 finds a solid evidence of backward technology transfers through buyer-provided training where a firm in a production chain is not only an information recipient from its downstream buyer but also an information sender to its upstream supplier. Local firms are also likely to provide training to their suppliers when they receive training from their buyers.

	(1)	(2)	(3)	(4)	(5)	(6)
Respondent firms in ASEAN	Whole	Local	Non-local	Whole	Local	Non-local
Incoming technology from buyer						
Main buyer gave training	0.368**	0.313**	0.435***	0.360***	0.328***	0.428***
	(0.040)	(0.058)	(0.061)	(0.041)	(0.059)	(0.065)
Distances to main supplier	(0.0.0)	(01000)	(01001)	(01011)	(0.00)	(0.000)
Log of distance to main supplier	-0.018	-0.026	0.023			
	(0.054)	(0.052)	(0.095)			
Square of distance to main supplier	0.001	0.004	-0.004			
Distances to main human	(0.006)	(0.005)	(0.010)			
Log of distance to main buyer	0.034	0.041	0.031			
Log of distance to main buyer	(0.054)	(0.041)	(0.100)			
Square of distance to main buyer	0.001	0.003	0.001			
	(0.006)	(0.005)	(0.010)			
Domestic vs. foreign supplier						
Domestic MNE supplier				0.082	0.070	0.071
				(0.075)	(0.084)	(0.128)
Domestic JV supplier				0.190***	0.117^{*}	0.251**
				(0.068)	(0.0/1)	(0.114)
Foreign supplier in ASEAN				0.279^{**}	(0.208)	$0.3/3^{***}$
Foreign supplier in East Asia				(0.109) 0.078	(0.191) 0 111	(0.123) 0.068
r orengin supprier in East Asia				(0.070)	(0.071)	(0.000)
Foreign supplier in EU or US				0.144	0.285*	0.193
				(0.123)	(0.173)	(0.174)
Foreign supplier in rest of the				0.005	-	0.231
.				(0.163)	(0.019)	(0.203)
Domestic vs. foreign buyer				0 170***	0 1 1 7	0.000
Domestic MNE buyer				$0.1/9^{***}$	(0.117)	0.320^{***}
Domestic IV buyer				-0.046	-0.007	-0.088
				(0.054)	(0.048)	(0.110)
Foreign buyer in ASEAN				0.099	0.014	0.185
				(0.084)	(0.103)	(0.130)
Foreign buyer in East Asia				-0.037	-	0.073
				(0.057)	(0.029)	(0.112)
Foreign buyer in EU or US				-0.029	(0.010)	0.015
Foreign buyer in rest of the world				(0.080)	(0.087)	(0.137) 0 314
i orengii buyer in rest or the world				(0.082)		(0.240)
<i>Buyer and supplier characteristics</i>				(0.002)		(0.210)
Capital tie with main buyer	-0.005	0.044	-0.055	-0.015	0.041	-0.089
	(0.049)	(0.063)	(0.078)	(0.049)	(0.061)	(0.079)
Log of firm size of main buyer	-0.054**	-0.014	-0.110**	-	-0.025	-0.141***
	(0.024)	(0.022)	(0.045)	(0.024)	(0.021)	(0.047)
Main buyer is doing R&D	0.135**	0.089**	0.207***	0.158***	0.110**	0.271***
	(0.040)	(0.045)	(0.064)	(0.038)	(0.046)	(0.063)
Capital tie with main supplier	0.264**	0.273**	0.297***	0.277***	0.264***	0.336***
	(0.061)	(0.089)	(0.083)	(0.062)	(0.090)	(0.090)
Log of firm size of main supplier	0.038	-0.005	0.079^{*}	0.037	0.001	0.083^{*}
Main supplier is doing R&D	(0.020)	-0.023	-0.024	-0.023	-0.024	-0.047
	(0.042)	(0.045)	(0.074)	(0.039)	(0.039)	(0.071)
Ν	807 [′]	423	384	807	417	384

Table 4: Impacts of Incoming Technology Transfers on Outgoing Technology Transfers

Note: All dependent variables are binary variables indicating whether the respondent firm in each country in ASEAN gave training to its main supplier. Reference group is domestic local buyer and local supplier. Other control variables are a capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, industry controls (seven manufacturing industry with reference industry), and country controls.
Source: ERIA Establishment Survey 2012.

4. Technology Transfer-driven Product Innovation

The next question is whether incoming and outgoing technology transfers encourage product innovation in developing economies. We believe that such investigation is important in order to start thinking of a broader question on the causes and consequences of knowledge acquisition, dissemination, and knowledge aggregation (Levitt, *et al.*, 2013) in buyer-supplier networks. We estimate the following equation:

$$\begin{aligned} Pr(Product\ innovation = 1)_i \\ &= \alpha + \delta_1(Buyer\ give\ training\)_i + \delta_2(Give\ training\ to\ supplier)_i \\ &+ \delta_B(Buyer\ give\ training)_i * (Give\ training\ to\ supplier)_i + \delta_X(covariates)_i \\ &+ v_i \qquad (4) \end{aligned}$$

where dependent variable (*product innovation*)_i is equal to one if firm *i* develops a new product with production partners, either its main buyer or supplier (for panel A of Table 5), or firm *i* designs a new product with its main buyer (for panel B of Table 5), otherwise zero. (*Buyer gave training*)_i is a dummy variable which is equal to one if firm *i*'s main buyer provides training to firm *i*, otherwise zero. (*Give training to supplier*)_i is a dummy variable which is equal to one if supplier firm *i*'s provides training to its upstream supplier, otherwise zero. The cross-term of (*Give training to supplier*)_i and (*Buyer gave training*)_i represents possible information aggregation by acquiring information through buyer-provided training and disseminating information through providing training to the supplier. (*covariates*)_i is a vector of firm-level observable characteristics capturing firm *i*'s capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, and industry controls as well as country controls, and v_i is an unobserved error.

Panel A of Table 5 checks impacts of information aggregation within a firm through incoming and outgoing technology transfers on developing a new product with production partners. Before introducing incoming and outgoing technology transfers, Column 1 of Panel A of Table 5 works with a simple question on how product development with production partners works with geography. It shows that the distance to the main buyer does not affect a decision of developing a new product with production partners. It also indicates that the probability of developing a new product with production partners has a U-shaped quadratic pattern with respect to the distance to the main supplier.

Let us move on to the impacts of information aggregation on product development. Column 2 of Panel A shows that the cross-term of "main buyer gave training" and "gave training to supplier" has positively correlated with developing a new product with production partners for the whole samples. Although the coefficients for "main buyer gave training" and "gave training to supplier" present negative signs, the individual effects turn out to be positive after summing up the coefficient for the crossterm. We can thus see for the whole samples that the in-formation aggregation through "main buyer gave training" and "gave training to supplier" works strongly on the probability of developing a new product with production partners. However, Column 3 finds that this is not true for local firms. There is no significant impact of the crossterm between incoming and outgoing technology transfers on product development for local samples. Column 4 of Panel A of Table 5 confirms that the effect of information aggregation comes only from the non-local firms.

Panel A. Dependent var.: Develop a new product with production partners	(1)	(2)	(3)	(4)
	Whole	Whole	Local	Non-local
Incoming and outgoing technologies				
Main buyer gave training		-0.064	-0.101**	0.002
Gave training to supplier		(0.045) -0.126**	(0.042) -0.024	(0.078) -0.229***
Main buyer gave training*Gave training to supplier		(0.053) 0.315***	(0.058) 0.102 (0.124)	(0.086) 0.424***
Distances to buyer and supplier		(0.094)	(0.124)	(0.117)
Log of distance to main buver (km)	0.022	0.045	-0.036	0.180**
Square of distance to main buyer	-0.003	-0.005	0.003	-0.018**
Log of distance to main supplier (km)	(0.005) -0.208***	(0.005) -0.200***	(0.006) -0.153***	(0.009) -0.241***
Square of distance to main supplier	(0.051) 0.019^{***}	(0.051) 0.019***	(0.056) 0.015***	(0.083) 0.022**
Ν	(0.005) 807	(0.005) 807	(0.006) 423	(0.008) 384
Panel B. Dependent var.: Develop a new product with main buyer	(5)	(6)	(7)	(8)
	Whole	Whole	Local	Non-local
Incoming and outgoing technologies				
Main buyer gave training		-0.005	0.011	-0.032
Gave training to supplier		(0.055) 0.243***	(0.069) 0.092	(0.095) 0.449***
Main buyer gave training*Gave training to supplier		(0.070) 0.255^{***} (0.098)	(0.096) 0.292* (0.152)	(0.099) 0.180 (0.139)
Distances to buyer and supplier		(0.020)	(0.102)	(0.127)
Log of distance to main buyer (km)	-0.226***	-0.150**	-0.044	-0.449***

Table 5: The Impacts of Incoming and Outgoing Technology Transfers on Product Innovation

	(0.063)	(0.064)	(0.086)	(0.115)
Square of distance to main buyer	0.022***	0.016**	0.008	0.044^{***}
	(0.006)	(0.006)	(0.008)	(0.011)
Log of distance to main supplier (km)	-0.296***	-0.275***	-0.268***	-0.371***
	(0.062)	(0.062)	(0.078)	(0.121)
Square of distance to main supplier	0.026***	0.024***	0.024***	0.034***
	(0.006)	(0.006)	(0.008)	(0.012)
Relationship with buyer and supplier				
Capital tie with main buyer		-0.095*	-0.040	-0.081
		(0.055)	(0.088)	(0.078)
Capital tie with main supplier		-0.177***	-0.170**	-0.267***
		(0.051)	(0.070)	(0.075)
Ν	807	807	423	384

Note: Other control variables are a capability score of process improvements, R&D activity, MNE dummy, JV dummy, firm age, firm size, OEM dummy, parts and components producer dummy, final goods producer dummies, industry controls (seven manufacturing industry with reference industry), and country controls. Source: ERIA Establishment Survey 2012.

Panel B of Table 5 also presents the impacts of incoming and outgoing technology transfers on designing a new product though we here focus on the innovative collaboration with the main buyer. Column 1 of Panel B of Table 5 shows that product development with the main buyer is also influenced by the geographic distance. Different from Column 1 of Panel A of Table 5, the probability of designing a new product with the main buyer presents a U-shaped quadratic pattern with respect to the distance to the main buyer, in addition to the distance to the main supplier.

We now look at impacts of information aggregation in the collaboration with the main buyer. In the case of the whole samples (Column 6 of Panel B), the cross-term of incoming and outgoing technology transfers shows positive association with product innovation with the main buyer. Compared with Panel A, however, we can see that information aggregation through "main buyer gave training" and "gave training to supplier" seems to encourage local firms to develop a new product with the main supplier (Column 7). On the other hand, non-local firms that gave training to the main supplier tend to develop a new product with the main supplier (Column 8). In sum, Panel B of Table 5 implies two things: (1) the probability of developing a new product with the main buyer presents a U-shaped quadratic pattern with respect to the distance to the main buyer and the main supplier; (2) information aggregation-driven product innovation seems to be effective for local firms.

5. Conclusion

This paper examines the pattern of technology transfers for product/process innovation in the form of buyer-provided training among local and non-local firms connected by domestic and international production networks. The buyer-supplier network data in Southeast Asia compiled by an ERIA research project provides a new opportunity to directly observe the supplier-buyer relationship as well as the existence of inter-firm provision of training as a form of technology transfers and investigate the geographical structure of knowledge aggregation among local and non-local firms. The data also contributes to find direct evidences for technology transfers through FDI that establishes affiliates in the host country and exports, the destination of which works as buyers.

Our empirical analysis based on the buyer-supplier network data provides the following findings: first, the probability of having training provided by the main buyer presents a U-shaped quadratic pattern with respect to the geographical distance between the respondent firms and the main buyers; i.e., the probability is high when the distance is either short or long. In the case of local firms, the geographical proximity to the main buyer seems to be particularly important in order to have training provided by the main buyer. Second, the training provision is especially likely for both local and non-local firms when the main buyer is a multinational located in the same country; it suggests that incoming FDI enhances technology transfers in domestic production networks. Non-local firms also tend to have training when the main buyer is located in ASEAN, East Asia, or the rest of the world, while such links do not work for local firms; exports in international production networks do not seem to work as a channel of technology transfers for local firms. Third, the probability of having training from the main buyer is high when the main buyer conducts R&D while the capital tie with the main buyer does not seem to enhance the probability. Fourth, both local and non-local firms that have training provided by their main buyers are likely to provide training to their main suppliers; i.e., production networks seem to work as a chain of technology transfers as well. Fifth, in the case of non-local firms, product innovation with production partners is more likely when they have upstream/downstream training; this suggests the link of innovation with technology acquisition, dissemination, and information aggregation. However, such links seem to be weaker in the case of local firms; upstream/ downstream training enhances product innovation working with the main buyers only at a weak statistical significance with large standard errors.

This paper has a limited scope but suggests a number of directions for further research. First, one strong message of this paper is the importance of direct observation of technology transfers. Although this paper concentrates on one particular form of technology transfer, i.e., buyer-provided training, in general we can think of various kinds of technology transfers or spillovers through a number of channels, some of which are certainly observable through hard-working micro data collection. We should explore this direction of research while certainly taking resource constraints into consideration. Second, production networks can also work as channels of technology

transfers, and the link between local firms in developing countries and multinationals must be explored further. Such studies are particularly applicable for developing countries like Southeast Asian countries in which production networks designed and operated mainly by multinationals have led industrialization. Third, geography matters for technology transfers. The importance of geographical proximity for technology transfers, particularly for local firms, suggests that proper designing of industrial agglomeration is crucial in developing countries. A number of Southeast Asian countries have aggressively utilized the mechanics of production networks and are now coming into the development stage of formulating industrial agglomeration that should effectively pick up positive agglomeration effects while avoiding congestion. In this context, the geographical structure of production networks, technology transfers, and upgrading innovation must continuously be investigated not only for academics but also for policy makers.

References

- Aitken, B. J. and A. E. Harrison (1999), 'Do Domestic Firms Benefit from Direct Foreign Investment? Evidence from Venezuela', *American Economic Review* 89(3), pp.605–18.
- Antràs, P., and E. Rossi-Hansberg (2009), 'Organizations and Trade', *Annual Review* of Economics, 1, pp.43–64
- Ando, M. and F. Kimura (2005), 'The Formation of International Production and Distribution Networks in East Asia', *NBER Chapters in International Trade in East Asia, NBER-East Asia Seminar on Economics*, 14, pp.177–216.
- Atalay, E., Hortacsu, A. and C. Syverson (2014), 'Vertical Integration and Input Flows', *American Economic Review*, 104(4), pp.1120–1148.
- Balsvik, R. (2011), 'Is Labor Mobility a Channel for Spillovers from Multinationals? Evidence from Norwegian Manufacturing', *The Review of Economics and Statistics*, 93, pp.285–297.
- Blalock, G., and P. J. Gertler (2008), 'Welfare Gains from Foreign Direct Investment through Technology Transfer to Local Suppliers', *Journal of International Economics*, 74(2), pp.402–21.
- Bloom, N., and J. Van Reenen (2007), 'Measuring and Explaining Management Practices across Firms and Countries', *Quarterly Journal of Economics*, 122(4), pp.351–408.
- Cohen W. M., and D. A. Levinthal (1990), 'Absorptive Capacity: A New Perspective on Learning and Innovation', *Administrative Science Quarterly*, 35(1),

pp.128–152.

- Conley, T. and C. Udry (2010), 'Learning about a new technology: pineapple in Ghana', *American Economic Review*, 100(1), pp.35–69.
- Gerschenberg, I. (1987), 'The Training and Spread of Managerial Know-How, A Comparative Analysis of Multinational and Other Firms in Kenya', *World Development*, 15(7), pp.931–939.
- Gorg, H., and E. Strobl (2005), 'Spillovers from Foreign Firms through Worker Mobility: An Empirical Investigation', *Scandinavian Journal of Economics*, 107, pp.693–709.
- Gorodnichenko, Y., J. Svejnar, and K. Terrel (2010), 'Globalization and Innovation in Emerging Markets', *American Economic Journal: Macroeconomics*, 2(2), pp. 194–226.
- Ichniowski, C., K. Shaw, and G. Prennushi (1997), 'The Effects of Human Resource Management Practices on Productivity: A Study of Steel Finishing Lines', *The American Economic Review*, 87(3), pp.291–313.
- Intarakumnerd, P. (2010), Fostering Production and Science & Technology Linkages to Stimulate Innovation in ASEAN, ERIA Research Project Report 2009, ERIA, No. 7–4.
- Javorcik, B. (2004), 'Does Foreign Direct Investment Increase the Productivity of Domestic Firms? In Search of Spillovers Through Backward Linkages', *American Economic Review*, 94(3), pp.605–27.
- Jones, R.W., and H. Kierzkowski (1990), 'The Role of Services in Production and International Trade: A Theoretical Framework', in: Jones R.W. and Krueger A. (eds.) The Political Economy of International Trade: Essays in Honour of Robert E. Baldwin, Oxford: Basil Blackwell, pp.31-48.
- Keller, W. (2004), 'International Technology Diffusion', *Journal of Economic Literature*, 42(3), pp.752–782.
- Keller, W. (2012), 'The Transfer of Knowledge across Countries', *NBER Reporter*, No.1, 2012.
- Kimura, F. and M. Ando (2005), 'Two-Dimensional Fragmentation in East Asia: Conceptual Framework and Empirics', *International Review of Economics and Finance*, 14(3), pp.317–348.
- Levitt, S. D., J. List, and C. Syverson (2013), 'Toward an Understanding of Learning by Doing: Evidence from an Automobile Assembly Plant', *Journal of Political Economy*, 121(4), pp.643–681.
- Machikita, T., and Y. Ueki (2013), 'Knowledge Transfer Channels to Vietnam for Process Improvement', *Management Decision*, 51(5), pp.954–972.
- Machikita, T., and Y. Ueki (2015), 'Measuring and Explaining Innovative Capability: Evidence from Southeast Asia', *Asian Economic Policy Review*, 10(1), pp.152–173.
- Markusen, J. R., and N. Trofimenko (2009), 'Teaching Locals New Tricks: Foreign Experts as a Channel of Knowledge Transfers', *Journal of Development*

Economics, 88(1), pp.120–131.

- Poole, J. (2013), 'Knowledge Transfers from Multinational to Domestic Firms: Evidence from Worker Mobility', *The Review of Economics and Statistics*, 95(2), pp.393–406.
- Stoyanov, A., and N. Zubanov (2012), 'Productivity Spillovers across Firms through Worker Mobility', American Economic Journal: Applied Economics, 4(2), pp.168–98.
- Truong, C, B. (2010), 'The Process of Improving Technology and the Perspective of Domestic Suppliers in the Motorbike Industry in Hanoi', *in:* Intarakumnerd, P. (ed.), *Fostering Production and Science & Technology Linkages to Stimulate Innovation in ASEAN*, Jakarta: ERIA, pp.71–305.

No.	Author(s)	Title	Year
2015-40	Fukunari KIMURA, Tomohiro MACHIKITA, and Yasushi UEKI	Technology Transfer in ASEAN Countries: Some Evidence from Buyer-Provided Training Network Data	May 2015
2015-39	Dionisius NARJOKO	AEC Blueprint Implementation Performance and Challenges: Services Liberalization	May 2015
2015-38	Kazunobu HAYAKAWA, Nuttawut LAKSANAPANYA KUL, Shujiro URATA	Measuring the Costs of FTA Utilization: Evidence from Transaction-level Import Data of Thailand	May 2015
2015-37	Kazunobu HAYAKAWA, Nuttawut LAKSANAPANYA KUL, Pisit PUAPAN, Sastra SUDSAWASD	Government Strategy and Support for Regional Trade Agreements: The Case of Thailand	May 2015
2015-36	Dionisius A. NARJOKO	AEC Blueprint Implementation Performance and Challenges: Non-Tariff Measures and Non-Tariff Barriers	May 2015
2015-35	Kazunobu HAYAKAWA, Tadashi ITO, and Fukunari KIMURA	Trade Creation Effects of Regional Trade Agreements: Tariff Reduction versus Non- tariff Barrier Removal	Apr 2015
2015-34	Kazunobu HAYAKAWA, Tadashi ITO	Tarrif Pass-through of the World-wide Trade: Empirical Evidence at Tarriff-line Level	Apr 2015
2015-33	Kazubobu HAYAKAWA, Nuttawut LAKSANAPNYAK UL, and Shujiro URATA	Firm-level Impact of Free Trade Agreements on Import Prices	Apr 2015
2015-32	Ponciano INTAL, Jr.	AEC Blueprint Implementation Performance and Challenges: Investment Liberalization	Apr 2015
2015-31	Emily Christi A. CABEGIN	The Challenge of China and the Role of Deepening ASEAN Integration for the Philippine Semiconductor Industry	Apr 2015
2015-30	Venkatachalam ANBUMOZHI, Alex BOWEN and	Market-Based Mechanisms to Promote Renewable Energy in Asia	Apr 2015

ERIA Discussion Paper Series

No.	Author(s)	Title	Year
	Puthusserikunnel Devasia JOSE		
2015-29	Venkatachalam ANBUMOZHI	Low Carbon Green Growth in Asia: What is the Scope for Regional Cooperation?	Apr 2015
2015-28	Tan LI and Larry D. QIU	Beyond Trade Creation: Free Trade Agreements and Trade Disputes	Mar 2015
2015-27	Mai Anh NGO	Exporting and Firm-Level Credit Constraints – Evidence from Ghana	Mar 2015
2015-26	Sunghoon CHUNG, Joonhyung LEE, Thomas OSANG	Did China Tire Safeguard Save U.S. Workers?	Mar 2015
2015-25	Esther Ann BØLER, Beata JAVORCIK, Karen Helene ULLTVEI-MOE	Globalization: A Woman's Best Friend? Exporters and the Gender Wage Gap	Mar 2015
2015-24	Tristan Leo Dallo AGUSTIN and Martin SCHRÖDER	The Indian Automotive Industry and the ASEAN Supply Chain Relations	Mar 2015
2015-23	Hideo KOBAYASHI and Yingshan JIN	The CLMV Automobile and Auto Parts Industry	Mar 2015
2015-22	Hideo KOBAYASHI	Current State and Issues of the Automobile and Auto Parts Industries in ASEAN	Mar 2015
2015-21	Yoshifumi FUKUNAGA	Assessing the Progress of ASEAN MRAs on Professional Services	Mar 2015
2015-20	Yoshifumi FUKUNAGA and Hikari ISHIDO	Values and Limitations of the ASEAN Agreement on the Movement of Natural Persons	Mar 2015
2015-19	Nanda NURRIDZKI	Learning from the ASEAN + 1 Model and the ACIA	Mar 2015
2015-18	Patarapong INTARAKUMNER D and Pun-Arj CHAIRATANA and Preeda CHAYANAJIT	Global Production Networks and Host-Site Industrial Upgrading: The Case of the Semiconductor Industry in Thailand	Feb 2015
2015-17	Rajah RASIAH and Yap Xiao SHAN	Institutional Support, Regional Trade Linkages and Technological Capabilities in the Semiconductor Industry in Singapore	Feb 2015
2015-16	Rajah RASIAH and Yap Xiao SHAN	Institutional Support, Regional Trade Linkages and Technological Capabilities in the Semiconductor Industry in Malaysia	Feb 2015
2015-15	Xin Xin KONG, Miao ZHANG and Santha Chenayah	China's Semiconductor Industry in Global Value Chains	Feb 2015

No.	Author(s)	Title	Year
	RAMU		
2015-14	Tin Htoo NAING and Yap Su FEI	Multinationals, Technology and Regional Linkages in Myanmar's Clothing Industry	Feb 2015
2015-13	Vanthana NOLINTHA and Idris JAJRI	The Garment Industry in Laos: Technological Capabilities, Global Production Chains and Competitiveness	Feb 2015
2015-12	Miao ZHANG, Xin Xin KONG, Santha Chenayah RAMU	The Transformation of the Clothing Industry in China	Feb 2015
2015-11	NGUYEN Dinh Chuc, NGUYEN Ngoc Anh, NGUYEN Ha Trang and NGUYEN Ngoc Minh	Host-site institutions, Regional Production Linkages and Technological Upgrading: A study of Automotive Firms in Vietnam	Feb 2015
2015-10	Pararapong INTERAKUMNER D and Kriengkrai TECHAKANONT	Intra-industry Trade, Product Fragmentation and Technological Capability Development in Thai Automotive Industry	Feb 2015
2015-09	Rene E. OFRENEO	Auto and Car Parts Production: Can the Philippines Catch Up with Asia	Feb 2015
2015-08	Rajah RASIAH, Rafat Beigpoor SHAHRIVAR, Abdusy Syakur AMIN	Host-site Support, Foreign Ownership, Regional Linkages and Technological Capabilites: Evidence from Automotive Firms in Indonesia	Feb 2015
2015-07	Yansheng LI, Xin Xin KONG, and Miao ZHANG	Industrial Upgrading in Global Production Networks: Te Case of the Chinese Automotive Industry	Feb 2015
2015-06	Mukul G. ASHER and Fauziah ZEN	Social Protection in ASEAN: Challenges and Initiatives for Post-2015 Vision	Feb 2015
2015-05	Lili Yan ING, Stephen MAGIERA, and Anika WIDIANA	Business Licensing: A Key to Investment Climate Reform	Feb 2015
2015-04	Gemma ESTRADA, James ANGRESANO, Jo Thori LIND, Niku MÄÄTÄNEN, William MCBRIDE, Donghyun PARK, Motohiro SATO, and Karin	Fiscal Policy and Equity in Advanced Economies: Lessons for Asia	Jan 2015

No.	Author(s)	Title	Year
	SVANBORG- SJÖVALL		
2015-03	Erlinda M. MEDALLA	Towards an Enabling Set of Rules of Origin for the Regional Comprehensive Economic Partnership	Jan 2015
2015-02	Archanun KOHPAIBOON and Juthathip JONGWANICH	Use of FTAs from Thai Experience	Jan 2015
2015-01	Misa OKABE	Impact of Free Trade Agreements on Trade in East Asia	Jan 2015
2014-26	Hikari ISHIDO	Coverage of Trade in Services under ASEAN+1 FTAs	Dec 2014
2014-25	Junianto James LOSARI	Searching for an Ideal International Investment Protection Regime for ASEAN + Dialogue Partners (RCEP): Where Do We Begin?	Dec 2014
2014-24	Dayong ZHANG and David C. Broadstock	Impact of International Oil Price Shocks on Consumption Expenditures in ASEAN and East Asia	Nov 2014
2014-23	Dandan ZHANG, Xunpeng SHI, and Yu SHENG	Enhanced Measurement of Energy Market Integration in East Asia: An Application of Dynamic Principal Component Analysis	Nov 2014
2014-22	Yanrui WU	Deregulation, Competition, and Market Integration in China's Electricity Sector	Nov 2014
2014-21	Yanfei LI and Youngho CHANG	Infrastructure Investments for Power Trade and Transmission in ASEAN+2: Costs, Benefits, Long-Term Contracts, and Prioritised Development	Nov 2014
2014-20	Yu SHENG, Yanrui WU, Xunpeng SHI, Dandan ZHANG	Market Integration and Energy Trade Efficiency: An Application of Malmqviat Index to Analyse Multi-Product Trade	Nov 2014
2014-19	Andindya BHATTACHARYA and Tania BHATTACHARYA	ASEAN-India Gas Cooperation: Redifining India's "Look East" Policy with Myanmar	Nov 2014
2014-18	Olivier CADOT, Lili Yan ING	How Restrictive Are ASEAN's RoO?	Sep 2014
2014-17	Sadayuki TAKII	Import Penetration, Export Orientation, and Plant Size in Indonesian Manufacturing	July 2014
2014-16	Tomoko INUI, Keiko ITO, and Daisuke MIYAKAWA	Japanese Small and Medium-Sized Enterprises' Export Decisions: The Role of Overseas Market Information	July 2014
2014-15	Han PHOUMIN and Fukunari KIMURA	Trade-off Relationship between Energy Intensity-thus energy demand- and Income Level: Empirical Evidence and Policy	June 2014

No.	Author(s)	Title	Year
		Implications for ASEAN and East Asia Countries	
2014-14	Cassey LEE	The Exporting and Productivity Nexus: Does Firm Size Matter?	May 2014
2014-13	Yifan ZHANG	Productivity Evolution of Chinese large and Small Firms in the Era of Globalisation	May 2014
2014-12	Valéria SMEETS, Sharon TRAIBERMAN, Frederic WARZYNSKI	Offshoring and the Shortening of the Quality Ladder:Evidence from Danish Apparel	May 2014
2014-11	Inkyo CHEONG	Korea's Policy Package for Enhancing its FTA Utilization and Implications for Korea's Policy	May 2014
2014-10	Sothea OUM, Dionisius NARJOKO, and Charles HARVIE	Constraints, Determinants of SME Innovation, and the Role of Government Support	May 2014
2014-09	Christopher PARSONS and Pierre-Louis Vézina	Migrant Networks and Trade: The Vietnamese Boat People as a Natural Experiment	May 2014
2014-08	Kazunobu HAYAKAWA and Toshiyuki MATSUURA	Dynamic Tow-way Relationship between Exporting and Importing: Evidence from Japan	May 2014
2014-07	DOAN Thi Thanh Ha and Kozo KIYOTA	Firm-level Evidence on Productivity Differentials and Turnover in Vietnamese Manufacturing	Apr 2014
2014-06	Larry QIU and Miaojie YU	Multiproduct Firms, Export Product Scope, and Trade Liberalization: The Role of Managerial Efficiency	Apr 2014
2014-05	Han PHOUMIN and Shigeru KIMURA	Analysis on Price Elasticity of Energy Demand in East Asia: Empirical Evidence and Policy Implications for ASEAN and East Asia	Apr 2014
2014-04	Youngho CHANG and Yanfei LI	Non-renewable Resources in Asian Economies: Perspectives of Availability, Applicability, Acceptability, and Affordability	Feb 2014
2014-03	Yasuyuki SAWADA and Fauziah ZEN	Disaster Management in ASEAN	Jan 2014
2014-02	Cassey LEE	Competition Law Enforcement in Malaysia	Jan 2014
2014-01	Rizal SUKMA	ASEAN Beyond 2015: The Imperatives for Further Institutional Changes	Jan 2014
2013-38	Toshihiro OKUBO, Fukunari KIMURA, Nozomu TESHIMA	Asian Fragmentation in the Global Financial Crisis	Dec 2013

No.	Author(s)	Title	Year
2013-37	Xunpeng SHI and Cecilya MALIK	Assessment of ASEAN Energy Cooperation within the ASEAN Economic Community	Dec 2013
2013-36	Tereso S. TULLAO, Jr. And Christopher James CABUAY	Eduction and Human Capital Development to Strengthen R&D Capacity in the ASEAN	Dec 2013
2013-35	Paul A. RASCHKY	Estimating the Effects of West Sumatra Public Asset Insurance Program on Short-Term Recovery after the September 2009 Earthquake	Dec 2013
2013-34	Nipon POAPONSAKORN and Pitsom MEETHOM	Impact of the 2011 Floods, and Food Management in Thailand	Nov 2013
2013-33	Mitsuyo ANDO	Development and Resructuring of Regional Production/Distribution Networks in East Asia	Nov 2013
2013-32	Mitsuyo ANDO and Fukunari KIMURA	Evolution of Machinery Production Networks: Linkage of North America with East Asia?	Nov 2013
2013-31	Mitsuyo ANDO and Fukunari KIMURA	What are the Opportunities and Challenges for ASEAN?	Nov 2013
2013-30	Simon PEETMAN	Standards Harmonisation in ASEAN: Progress, Challenges and Moving Beyond 2015	Nov 2013
2013-29	Jonathan KOH and Andrea Feldman MOWERMAN	Towards a Truly Seamless Single Windows and Trade Facilitation Regime in ASEAN Beyond 2015	Nov 2013
2013-28	Rajah RASIAH	Stimulating Innovation in ASEAN Institutional Support, R&D Activity and Intelletual Property Rights	Nov 2013
2013-27	Maria Monica WIHARDJA	Financial Integration Challenges in ASEAN beyond 2015	Nov 2013
2013-26	Tomohiro MACHIK ITA and Yasushi UE KI	Who Disseminates Technology to Whom, How, and Why: Evidence from Buyer-Seller Business Networks	Nov 2013
2013-25	Fukunari KIMURA	Reconstructing the Concept of "Single Market a Production Base" for ASEAN beyond 2015	Oct 2013
2013-24	Olivier CADOT Ernawati MUNADI Lili Yan ING	Streamlining NTMs in ASEAN: The Way Forward	Oct 2013
2013-23	Charles HARVIE, Dionisius NARJOK O, Sothea OUM	Small and Medium Enterprises' Access to Finance: Evidence from Selected Asian Economies	Oct 2013
2013-22	Alan Khee-Jin TAN	Toward a Single Aviation Market in ASEAN: Regulatory Reform and Industry Challenges	Oct 2013
2013-21	Hisanobu SHISHID O, Shintaro SUGIYAM	Moving MPAC Forward: Strengthening Public-Private Partnership, Improving Project Portfolio and in Search of Practical Financing	Oct 2013

No.	Author(s)	Title	Year
	A,Fauziah ZEN	Schemes	
2013-20	Barry DESKER, Mely CABALLERO- ANTHONY, Paul TENG	Thought/Issues Paper on ASEAN Food Security: Towards a more Comprehensive Framework	Oct 2013
2013-19	Toshihiro KUDO, Satoru KUMAGAI, So UMEZAKI	Making Myanmar the Star Growth Performer in ASEAN in the Next Decade: A Proposal of Five Growth Strategies	Sep 2013
2013-18	Ruperto MAJUCA	Managing Economic Shocks and Macroeconomic Coordination in an Integrated Region: ASEAN Beyond 2015	Sep 2013
2013-17	Cassy LEE and Yoshifumi FUKUNAGA	Competition Policy Challenges of Single Market and Production Base	Sep 2013
2013-16	Simon TAY	Growing an ASEAN Voice? : A Common Platform in Global and Regional Governance	Sep 2013
2013-15	Danilo C. ISRAEL and Roehlano M. BRIONES	Impacts of Natural Disasters on Agriculture, Food Security, and Natural Resources and Environment in the Philippines	Aug 2013
2013-14	Allen Yu-Hung LAI and Seck L. TAN	Impact of Disasters and Disaster Risk Management in Singapore: A Case Study of Singapore's Experience in Fighting the SARS Epidemic	Aug 2013
2013-13	Brent LAYTON	Impact of Natural Disasters on Production Networks and Urbanization in New Zealand	Aug 2013
2013-12	Mitsuyo ANDO	Impact of Recent Crises and Disasters on Regional Production/Distribution Networks and Trade in Japan	Aug 2013
2013-11	Le Dang TRUNG	Economic and Welfare Impacts of Disasters in East Asia and Policy Responses: The Case of Vietnam	Aug 2013
2013-10	Sann VATHANA, Sothea OUM, Ponhrith KAN, Colas CHERVIER	Impact of Disasters and Role of Social Protection in Natural Disaster Risk Management in Cambodia	Aug 2013
2013-09	Sommarat CHANTARAT, Krirk PANNANGPETCH, Nattapong PUTTANAPONG, Preesan RAKWATIN, and Thanasin TANOMPONGPHA	Index-Based Risk Financing and Development of Natural Disaster Insurance Programs in Developing Asian Countries	Aug 2013

No.	Author(s)	Title	Year
	NDH		
2013-08	Ikumo ISONO and Satoru KUMAGAI	Long-run Economic Impacts of Thai Flooding: Geographical Simulation Analysis	July 2013
2013-07	Yoshifumi FUKUNAGA and Hikaru ISHIDO	Assessing the Progress of Services Liberalization in the ASEAN-China Free Trade Area (ACFTA)	May 2013
2013-06	Ken ITAKURA, Yoshifumi FUKUNAGA, and Ikumo ISONO	A CGE Study of Economic Impact of Accession of Hong Kong to ASEAN-China Free Trade Agreement	May 2013
2013-05	Misa OKABE and Shujiro URATA	The Impact of AFTA on Intra-AFTA Trade	May 2013
2013-04	Kohei SHIINO	How Far Will Hong Kong's Accession to ACFTA will Impact on Trade in Goods?	May 2013
2013-03	Cassey LEE and Yoshifumi FUKUNAGA	ASEAN Regional Cooperation on Competition Policy	Apr 2013
2013-02	Yoshifumi FUKUNAGA and Ikumo ISONO	Taking ASEAN+1 FTAs towards the RCEP: A Mapping Study	Jan 2013
2013-01	Ken ITAKURA	Impact of Liberalization and Improved Connectivity and Facilitation in ASEAN for the ASEAN Economic Community	Jan 2013
2012-17	Sun XUEGONG, Guo LIYAN, Zeng ZHENG	Market Entry Barriers for FDI and Private Investors: Lessons from China's Electricity Market	Aug 2012
2012-16	Yanrui WU	Electricity Market Integration: Global Trends and Implications for the EAS Region	Aug 2012
2012-15	Youngho CHANG, Yanfei LI	Power Generation and Cross-border Grid Planning for the Integrated ASEAN Electricity Market: A Dynamic Linear Programming Model	Aug 2012
2012-14	Yanrui WU, Xunpeng SHI	Economic Development, Energy Market Integration and Energy Demand: Implications for East Asia	Aug 2012
2012-13	Joshua AIZENMAN, Minsoo LEE, and Donghyun PARK	The Relationship between Structural Change and Inequality: A Conceptual Overview with Special Reference to Developing Asia	July 2012
2012-12	Hyun-Hoon LEE, Minsoo LEE, and Donghyun PARK	Growth Policy and Inequality in Developing Asia: Lessons from Korea	July 2012
2012-11	Cassey LEE	Knowledge Flows, Organization and Innovation: Firm-Level Evidence from	June 2012

No.	Author(s)	Title	Year
		Malaysia	
2012-10	Jacques MAIRESSE, Pierre MOHNEN, Yayun ZHAO, and Feng ZHEN	Globalization, Innovation and Productivity in Manufacturing Firms: A Study of Four Sectors of China	June 2012
2012-09	Ari KUNCORO	Globalization and Innovation in Indonesia: Evidence from Micro-Data on Medium and Large Manufacturing Establishments	June 2012
2012-08	Alfons PALANGKARAYA	The Link between Innovation and Export: Evidence from Australia's Small and Medium Enterprises	June 2012
2012-07	Chin Hee HAHN and Chang-Gyun PARK	Direction of Causality in Innovation-Exporting Linkage: Evidence on Korean Manufacturing	June 2012
2012-06	Keiko ITO	Source of Learning-by-Exporting Effects: Does Exporting Promote Innovation?	June 2012
2012-05	Rafaelita M. ALDABA	Trade Reforms, Competition, and Innovation in the Philippines	June 2012
2012-04	Toshiyuki MATSUURA and Kazunobu HAYAKAWA	The Role of Trade Costs in FDI Strategy of Heterogeneous Firms: Evidence from Japanese Firm-level Data	June 2012
2012-03	Kazunobu HAYAKAWA, Fukunari KIMURA, and Hyun-Hoon LEE	How Does Country Risk Matter for Foreign Direct Investment?	Feb 2012
2012-02	Ikumo ISONO, Satoru KUMAGAI, Fukunari KIMURA	Agglomeration and Dispersion in China and ASEAN: A Geographical Simulation Analysis	Jan 2012
2012-01	Mitsuyo ANDO and Fukunari KIMURA	How Did the Japanese Exports Respond to Two Crises in the International Production Network?: The Global Financial Crisis and the East Japan Earthquake	Jan 2012
2011-10	Tomohiro MACHIKITA and Yasushi UEKI	Interactive Learning-driven Innovation in Upstream-Downstream Relations: Evidence from Mutual Exchanges of Engineers in Developing Economies	Dec 2011
2011-09	Joseph D. ALBA, Wai-Mun CHIA, and Donghyun PARK	Foreign Output Shocks and Monetary Policy Regimes in Small Open Economies: A DSGE Evaluation of East Asia	Dec 2011

No.	Author(s)	Title	Year
2011-08	Tomohiro MACHIKITA and Yasushi UEKI	Impacts of Incoming Knowledge on Product Innovation: Econometric Case Studies of Technology Transfer of Auto-related Industries in Developing Economies	Nov 2011
2011-07	Yanrui WU	Gas Market Integration: Global Trends and Implications for the EAS Region	Nov 2011
2011-06	Philip Andrews- SPEED	Energy Market Integration in East Asia: A Regional Public Goods Approach	Nov 2011
2011-05	Yu SHENG, Xunpeng SHI	Energy Market Integration and Economic Convergence: Implications for East Asia	Oct 2011
2011-04	Sang-Hyop LEE, Andrew MASON, and Donghyun PARK	Why Does Population Aging Matter So Much for Asia? Population Aging, Economic Security and Economic Growth in Asia	Aug 2011
2011-03	Xunpeng SHI, Shinichi GOTO	Harmonizing Biodiesel Fuel Standards in East Asia: Current Status, Challenges and the Way Forward	May 2011
2011-02	Hikari ISHIDO	Liberalization of Trade in Services under ASEAN+n : A Mapping Exercise	May 2011
2011-01	Kuo-I CHANG, Kazunobu HAYAKAWA Toshiyuki MATSUURA	Location Choice of Multinational Enterprises in China: Comparison between Japan and Taiwan	Mar 2011
2010-11	Charles HARVIE, Dionisius NARJOKO, Sothea OUM	Firm Characteristic Determinants of SME Participation in Production Networks	Oct 2010
2010-10	Mitsuyo ANDO	Machinery Trade in East Asia, and the Global Financial Crisis	Oct 2010
2010-09	Fukunari KIMURA Ayako OBASHI	International Production Networks in Machinery Industries: Structure and Its Evolution	Sep 2010
2010-08	Tomohiro MACHIKITA, Shoichi MIYAHARA, Masatsugu TSUJI, and Yasushi UEKI	Detecting Effective Knowledge Sources in Product Innovation: Evidence from Local Firms and MNCs/JVs in Southeast Asia	Aug 2010
2010-07	Tomohiro MACHIKITA, Masatsugu TSUJI, and Yasushi UEKI	How ICTs Raise Manufacturing Performance: Firm-level Evidence in Southeast Asia	Aug 2010
2010-06	Xunpeng SHI	Carbon Footprint Labeling Activities in the East Asia Summit Region: Spillover Effects to Less Developed Countries	July 2010
2010-05	Kazunobu HAYAKAWA, Fukunari KIMURA, and	Firm-level Analysis of Globalization: A Survey of the Eight Literatures	Mar 2010

No.	Author(s)	Title	Year
	Tomohiro MACHIKITA		
2010-04	Tomohiro MACHIKITA and Yasushi UEKI	The Impacts of Face-to-face and Frequent Interactions on Innovation: Upstream-Downstream Relations	Feb 2010
2010-03	Tomohiro MACHIKITA and Yasushi UEKI	Innovation in Linked and Non-linked Firms: Effects of Variety of Linkages in East Asia	Feb 2010
2010-02	Tomohiro MACHIKITA and Yasushi UEKI	Search-theoretic Approach to Securing New Suppliers: Impacts of Geographic Proximity for Importer and Non-importer	Feb 2010
2010-01	Tomohiro MACHIKITA and Yasushi UEKI	Spatial Architecture of the Production Networks in Southeast Asia: Empirical Evidence from Firm-level Data	Feb 2010
2009-23	Dionisius NARJOKO	Foreign Presence Spillovers and Firms' Export Response: Evidence from the Indonesian Manufacturing	Nov 2009
2009-22	Kazunobu HAYAKAWA, Daisuke HIRATSUKA, Kohei SHIINO, and Seiya SUKEGAWA	Who Uses Free Trade Agreements?	Nov 2009
2009-21	Ayako OBASHI	Resiliency of Production Networks in Asia: Evidence from the Asian Crisis	Oct 2009
2009-20	Mitsuyo ANDO and Fukunari KIMURA	Fragmentation in East Asia: Further Evidence	Oct 2009
2009-19	Xunpeng SHI	The Prospects for Coal: Global Experience and Implications for Energy Policy	Sept 2009
2009-18	Sothea OUM	Income Distribution and Poverty in a CGE Framework: A Proposed Methodology	Jun 2009
2009-17	Erlinda M. MEDALLA and Jenny BALBOA	ASEAN Rules of Origin: Lessons and Recommendations for the Best Practice	Jun 2009
2009-16	Masami ISHIDA	Special Economic Zones and Economic Corridors	Jun 2009
2009-15	Toshihiro KUDO	Border Area Development in the GMS: Turning the Periphery into the Center of Growth	May 2009
2009-14	Claire HOLLWEG and Marn-Heong WONG	Measuring Regulatory Restrictions in Logistics Services	Apr 2009
2009-13	Loreli C. De DIOS	Business View on Trade Facilitation	Apr 2009
2009-12	Patricia SOURDIN and Richard POMFRET	Monitoring Trade Costs in Southeast Asia	Apr 2009

No.	Author(s)	Title	Year
2009-11	Philippa DEE and Huong DINH	Barriers to Trade in Health and Financial Services in ASEAN	Apr 2009
2009-10	Sayuri SHIRAI	The Impact of the US Subprime Mortgage Crisis on the World and East Asia: Through Analyses of Cross-border Capital Movements	Apr 2009
2009-09	Mitsuyo ANDO and Akie IRIYAMA	International Production Networks and Export/Import Responsiveness to Exchange Rates: The Case of Japanese Manufacturing Firms	Mar 2009
2009-08	Archanun KOHPAIBOON	Vertical and Horizontal FDI Technology Spillovers:Evidence from Thai Manufacturing	Mar 2009
2009-07	Kazunobu HAYAKAWA, Fukunari KIMURA, and Toshiyuki MATSUURA	Gains from Fragmentation at the Firm Level: Evidence from Japanese Multinationals in East Asia	Mar 2009
2009-06	Dionisius A. NARJOKO	Plant Entry in a More LiberalisedIndustrialisationProcess: An Experience of Indonesian Manufacturing during the 1990s	Mar 2009
2009-05	Kazunobu HAYAKAWA, Fukunari KIMURA, and Tomohiro MACHIKITA	Firm-level Analysis of Globalization: A Survey	Mar 2009
2009-04	Chin Hee HAHN and Chang-Gyun PARK	Learning-by-exporting in Korean Manufacturing: A Plant-level Analysis	Mar 2009
2009-03	Ayako OBASHI	Stability of Production Networks in East Asia: Duration and Survival of Trade	Mar 2009
2009-02	Fukunari KIMURA	The Spatial Structure of Production/Distribution Networks and Its Implication for Technology Transfers and Spillovers	Mar 2009
2009-01	Fukunari KIMURA and Ayako OBASHI	International Production Networks: Comparison between China and ASEAN	Jan 2009
2008-03	Kazunobu HAYAKAWA and Fukunari KIMURA	The Effect of Exchange Rate Volatility on International Trade in East Asia	Dec 2008
2008-02	Satoru KUMAGAI, Toshitaka GOKAN, Ikumo ISONO, and Souknilanh KEOLA	Predicting Long-Term Effects of Infrastructure Development Projects in Continental South East Asia: IDE Geographical Simulation Model	Dec 2008
2008-01	Kazunobu HAYAKAWA, Fukunari KIMURA, and Tomohiro MACHIKITA	Firm-level Analysis of Globalization: A Survey	Dec 2008